Author's response to reviews

Title: Syndrome of Inappropriate Antidiuretic Hormone (ADH) Associated with Ectopic ADH-secreting Gastric Adenocarcinoma: a case report

Authors:

Kyungo Hwang (hkomanse@hanmail.net)
Dae-Hong Jeon (eoghd82@naver.com)
Ha Nee Jang (asaku@naver.com)
Eun Jin Bae (delight7607@naver.com)
Jong Sil Lee (jongsil25@dreamwiz.com)
Hyun Seop Cho (mars36@hanmail.net)
Se-Ho Chang (shchang@gnu.ac.kr)
Dong Jun Park (drpdj@korea.com)

Version: 3 Date: 26 March 2014

Author's response to reviews:

1. Please include all the co-authors' email addresses in the Title page.
   - We described all the co-authors' email addresses in the Title page

2. Please reposition the Keywords section after the Abstract section.
   - We repositioned the Keywords section after the Abstract section.

3. Please include the Ethnicity of the patient in the Abstract and Case presentation sections.
   - We described the Ethnicity of the patient in the Abstract and Case presentation sections.

4. Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.
   - Abbreviations in my manuscripts was described after the Conclusions section.

5. Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.
   - We have no the Acknowledgements